%&latex
%
% vjmp-template.tex  - Vassar Journal of Modern Physics template file
%
% @(#) $Id: vjmp-template.tex,v 1.1 2004/02/14 19:17:10 myers Exp myers $
%
%  This file is a template for papers produced for Physics 201
%  (Modern Physics Laboratory) at Vassar College using LaTeX & REVTeX 4.
%
% Lines like this that begin with `%' are comments and are ignored by LaTeX.


% LaTeX documents always have a \documentclass{} to set the output format.
% Adding options to controll the formatting and layout:
%  * The 'preprint' format is suitable for submission for grading
%  * Add 'draft' option to mark overfull boxes with black boxes and 
%      to only show included graphics as empty boxes of the correct size.      
%  * The 'twocolumn' format gives a double column layout like the
%      Physical Review Journals. Choose pra, prb, prc, prd, pre, prl,
%      prstab, or rmp for a particular journal style.
%  * Include 'twoside' for odd/even page differences.
%  * Add 'floatfix' if there are problems with floating diagrams.

\documentclass[aps,prl,preprint]{revtex4}
%\documentclass[aps,prl,twocolumn,draft,floatfix]{revtex4}
%\documentclass[aps,prl,twocolumn,twoside]{revtex4}


%% Special packages to use for graphics and such:
\usepackage{graphicx}% so you can include EPS figure files (see example below)


%% Uncomment these to use vjmp.sty style file:
%
%\usepackage{vjmp}
%  \def\theAuthor{Anne Author}  
%  \def\theVolume{2}
%  \def\pageRange{47-52}
%  \def\theYear{2004}


%% Every LaTeX document begins with \begin{document} and ends with \end{document}

\begin{document}
  \date{December 7, 1941}
%  \pagestyle{vjmp}
%  \setcounter{page}{47}

%======================================================================%
% Title material:

% Use the \preprint command to place your local institutional report
% number in the upper righthand corner of the title page in preprint mode.
\preprint{VC-MPL-2004-xx}

% Title of paper:
\title{Put the title of your paper here}

\author{First Author here}
\email[]{first-author-email@vassar.edu}

\author{Second Author here}
\email[]{second-author-email@vassar.edu}

% \affiliation command applies to all authors since the last \affiliation 
% command. This should follow the other information (email, etc...)
\affiliation{Modern Physics Laboratory \\
        Department of Physics and Astronomy \\ 
        Vassar College, 
        Poughkeepsie, New York  12604 USA}


\begin{abstract}
Text of your abstract goes here.
\end{abstract}


\thanks{Any acknowledgements go here.  For example, you can
   Thank your URSI or REU advisor here.}


% \maketitle must follow title, authors, abstract, \pacs, and \keywords
% It actually makes the title page (or title material on first page).
\maketitle


%======================================================================%
% Body of paper here - Use proper section commands
% Put \label in argument of \section for cross-referencing.


\section{Introduction   \label{intro}}
Introductory section here...

\section{Background and Theory}
Another section could go here...

Here is an example of a 'displayed' equation which can be refered to in
the text via a label:
\begin{equation}
  \sigma_i\sigma_j = i\sigma_k \qquad  \text{($i,j,k $  cyclic)}
\label{eq:pauli}
\end{equation}
You would refer to this in the text as Eq.~\ref{eq:bohr} and the 
number will automatcially come out correct


\section{Apparatus and Methods}
Another section...

Here is an example of how to include a figure using the file 
FigFile.eps and including a caption:

\begin{figure}
\includegraphics[width=\columnwidth]{FigFile}
\caption{Caption text goes here, with a label so that you can
        refer to the figure in the text by label
        \label{fig:some-figure-tag}}      %  the "fig:" is just a convention
\end{figure}

You would refer to this in the text as Fig.~\ref{fig:some-figure-tag}
and the right figure number would be used automatically.


\section{Data and Analysis}

\subsection{Presentation of Data}
Use subsections as appropriate...

\subsection{Calculations}
Use subsections to separate out different aspects of a section...

Here is an example of a table.  The caption comes first for a table so
that it appears above the table.  The caption comes second for a figure
so that the figure caption is below the drawing or figure.

\begin{table}%
  \caption{caption goes here, with a label
          \label{tab:ATT}}
  \begin{ruledtabular}
    \begin{tabular}{rcr}% r=right justify, l=left justify, c=centered
     Year & Price & Dividend \\
    \hline
     1971 & 41-54 & \$2.60   \\
     ~~~2 & 41-54 & \$2.70   \\
     ~~~3 & 46-55 & \$2.87   \\
     ~~~4 & 40-53 & \$3.24   \\
     ~~~5 & 45-52 & \$3.40   \\
     ~~~6 & 51-59 &  ~~.95   \\
    \end{tabular}
  \end{ruledtabular}
\end{table}


\subsection{Error Analysis}
Yet another subsection...


Cite references to journal articles, books etc like so:
You can go to a well recognized source, such as the Particle
Databook,\cite{PDG94} and look up the accepted value of this quantity.

There should be an entry in your .bib file called `PDG94' for the
item you cited.
To cite several works at once just include them all in the citation,
as in ...or from Bevington's
book\cite{Bevington1969,Bevington1992}) to determine what the
likely statistical uncertainty is in your final value.

Unpublished information which has been made available to you personally
by someone else should be referenced as ``private
communication''.\cite{Jamie}

References to web pages should include the author's name, the title of
the web page (which generally appears in the ``title bar'' at the very
top of the browser window, the date the page was produced, and the URL.
You may need to use the browser's ``get information'' feature to find
out when a page was originally produced.  Since the web is constantly
changing, it is also suggested that you note the date on which you
personally visited the page.

It may be difficult to determine the author of a web page, but doing so
is important.
Is it someone with a reliable reputation, or just an anonymous
high school student with no real authority?
Before you cite a source on the web, you might consider whether you
can get the same information from a more reliable source.
Although the web is useful and provides a lot of valid information, many
scholars are not inspired by the phrase ``{\em I read it on the
Internet.}''


%======================================================================

\section{Summary and Conclusions}
Always end your paper with a summary or conclusions section, don't just
leave things hanging...



% If you have acknowledgments, this puts in the proper section head when
% apropriate.

\bigskip
\begin{acknowledgments}
This creates an entire section for acknowlegements, which you
may or may not need.
\end{acknowledgments}


% Specify following sections are appendices. Use \appendix* if there
% only one appendix.
\appendix*
\section{Appendix:  title goes here}
Put text of an appendix here, if there is one...


% This creates the References section by reading in the file
% vjmp-template.bbl, which is created from the files vjmp-template.bib
% and vjmp-template.aux by the bibtex program.   This all assumes that
% your manuscript file is called vjmp-template.tex , so you should
% change the name to that of your own manuscript file.
 
\bibliography{vjmp-template}

% Don't forget to end with this...

\end{document}

%%EOF - end of template file

